Author’s response to reviews

Title: Who gets a family physician through centralized waiting lists?

Authors:

Mylaine Breton (mylaine.breton@usherbrooke.ca)
Astrid Brouselle (astrid.brouselle@usherbrooke.ca)
Antoine Boivin (antoie.boivin@usherbrooke.ca)
Danièle Roberge (daniele.roberge@usherbrooke.ca)
Raynald Pineault (rpineault@santepub-mtl.qc.ca)
Djamal Berbiche (djamal.berbiche@usherbrooke.ca)

Version: 2 Date: 25 September 2014

Author’s response to reviews:

Dear editor,

As request, we made some change;
1) add a sentence in the methods section on ethical approval
2) In the abstract, we use the classic section; we delete objective section and we did not use abbreviation
3) add line numbering and did not use page break
4) we are not using supporting data section
5) Add some section according to STROBE guidelines.

Bests regards
Mylaine Breton